Author's response to reviews

Title: Clinical approaches to treating papillary squamous cell carcinoma of the uterine cervix

Authors:

Michikazu Nagura (mnagura@kuhp.kyoto-u.ac.jp)
Masafumi Koshiyama (koshiyamam@nifty.com)
Noriomi Matsumura (noriomi@kuhp.kyoto-u.ac.jp)
Aki Kido (akikido@kuhp.kyoto-u.ac.jp)
Tsukasa Baba (babatsu@kuhp.kyoto-u.ac.jp)
Kaoru Abiko (kaoruvc@kuhp.kyoto-u.ac.jp)
Junzo Hamanishi (jnkhmns@kuhp.kyoto-u.ac.jp)
Ken Yamaguchi (soulken@kuhp.kyoto-u.ac.jp)
Yoshiki Mikami (mika@kuhp.kyoto-u.ac.jp)
Ikuo Konishi (konishi@kuhp.kyoto-u.ac.jp)

Version: 7 Date: 1 October 2014

Author's response to reviews: see over
Dear Dr. Koshiyama,

You have stated in your manuscript that 16 of the SCC or adenosquamous carcinoma exhibited relatively poor prognoses, in such cases if patients were deceased, please can you clarify how you obtained consent to publish. Since we have concerns regarding anonymity of patients, please anonymize both additional tables 1 and 2 by removing age.

Answer:

>>> Yes, I removed patients’ ages from Table 1 and 2.